E&CE318, Winter 2002

Text Book ffl F. G. Stremler, Introduction to communication systems, Third edition, Addison-Wesley, 1992, Call number: TK5101.S75

References ffl A. B. Carlson, Communication systems : an introduction to signals and noise in electrical communication, Third edition, McGraw-Hill , 1986, Call number: TK5102.5.C3 1986

Timetable ffl Lectures: 10:30-11:20MWF MC-2054 ffl Tutorial: 10:30-11:20Th DWE-2527, First tutorial: Thursday, Jan. 10th Marking: 25% Lab, 25% Midterm and 50% Final.

Lab. Technologist: Kevin Luscott, E2 3355, X6881, kluscott@engmail Teaching Assistant (lab): Hadi Baligh, DC 2551A, X2327, hadi@cst Teaching Assistant (tutorial): Farzaneh Kohandani, DC 2553, X2327, farzaneh@cst Instructor: A. K. Khandani, DC2740, X5324, khandani@cst